Author's response to reviews

Title: Influences on recruitment to randomised controlled trials in mental health settings in England: a national cross-sectional survey of clinical studies officers working for the Mental Health Research Network

Authors:

Rohan Borschmann (rohan.borschmann@kcl.ac.uk)
Sue Patterson (susan.patterson@health.qld.gov.au)
Dilkushi Poovendran (d.poovendran@imperial.ac.uk)
Danielle Wilson (d.wilson@imperial.ac.uk)
Tim Weaver (t.weaver@imperial.ac.uk)

Version: 2 Date: 11 October 2013

Author's response to reviews: see over
Dear Ms. Pura,

**RE: Influences on recruitment to randomised controlled trials in mental health settings in England: a national cross-sectional survey of clinical studies officers working for the Mental Health Research Network (MS: 2111595925108459).**

As requested, I have uploaded a copy of the survey instrument used in the above study. If you require further revisions we will be happy to consider them, thank you.

We look forward to hearing from you.

Yours sincerely,

Dr. Rohan Borschmann
Corresponding author